Author's response to reviews

Title: Randomized clinical trial comparing percutaneous closure of patent foramen ovale (PFO) using the Amplatzer PFO Occluder with medical treatment in patients with cryptogenic embolism (PC-Trial): Rationale and design.

Authors:

Ahmed A. Khattab (ahmed.khattab@insel.ch)
Stephan Windecker (stephan.windecker@insel.ch)
Peter Juni (juni@ispm.unibe.ch)
David Hildick-Smith (david@gmail.com)
Dariusz Dudek (dudek@gmail.com)
Henning R. Andersen (andersen@gmail.com)
Reda Ibrahim (ibrahim@gmail.com)
Gerhard Schuler (schuler@hotmail.com)
Antony S. Walton (antony@hotmail.com)
Andreas Wahl (andreas.wahl@insel.ch)
Heinrich P. Mattle (heinrich@mail.com)
Bernhard Meier (bernhardmeier@insel.ch)

Version: 4 Date: 12 January 2011

Author's response to reviews:

We changed figure 1 to be table 1.
Tables 1-4 were kept in the same format but are now indicated as appendices.
Reference 3 was added into the text.
We added the sentence that the authors have read and approved the final manuscript.
We removed the additional files.